Author’s response to reviews

Title: Are preterm birth and intra-uterine growth restriction more common in Western Australian children of immigrant backgrounds? A population based data linkage study.

Authors:
Ifrah Abdullahi (ifrah.abdullahi@telethonkids.org.au)
Kingsley Wong (kingsley.wong@telethonkids.org.au)
Emma Glasson (emma.glasson@telethonkids.org.au)
Raewyn Mutch (raewyn.mutch@health.wa.gov.au)
Nick de Klerk (nick.deklerk@telethonkids.org.au)
Jenny Downs (jenny.downs@telethonkids.org.au)
Sarah Cherian (sarah.cherian@health.wa.gov.au)
Helen Leonard (helen.leonard@telethonkids.org.au)

Version: 4 Date: 26 Jul 2019

Author’s response to reviews:

PRCH-D-17-00848R3

Are preterm birth and intra-uterine growth restriction more common in Western Australian children of immigrant backgrounds? A population based data linkage study.

Ifrah Abdullahi, BSc(Hons); Kingsley Wong, MBBS, MPH; Emma Glasson, PhD; Raewyn Mutch, MBBS(Hons), FRACP, PhD; Nick de Klerk, PhD; Jenny Downs, PhD; Sarah Cherian, MBBS(Hons), FRACP, PhD; Helen Leonard, MBChB, MPH

BMC Pregnancy and Childbirth

Dear Dr Leonard,

Your manuscript "Are preterm birth and intra-uterine growth restriction more common in Western Australian children of immigrant backgrounds? A population based data linkage study." (PRCH-D-17-00848R3) has been assessed by our reviewers. Based on these reports, and my own assessment as Assistant Editor, I am pleased to inform you that it is potentially acceptable for
publication in BMC Pregnancy and Childbirth, once you have carried out some essential revisions.

Once you have made the necessary corrections, please submit a revised manuscript online at:

https://www.editorialmanager.com/prch/

If you have forgotten your password, please use the 'Send Login Details' link on the login page at https://www.editorialmanager.com/prch/. For security reasons, your password will be reset.

Please also ensure that your revised manuscript conforms to the journal style, which can be found at the Submission Guidelines on the journal homepage.

A decision will be made once we have received your revised manuscript, which we expect by 17 Aug 2019.

Please note that you will not be able to add, remove, or change the order of authors once the editor has accepted your manuscript for publication. Any proposed changes to the authorship must be requested during peer-review, and adhere to our criteria for authorship as outlined in BioMed Central's policies. To request a change in authorship, please download the 'Request for change in authorship form' which can be found here - http://www.biomedcentral.com/about/editorialpolicies#authorship. Please note that incomplete forms will be rejected. Your request will be taken into consideration by the editor, and you will be advised whether any changes will be permitted. Please be aware that we may investigate, or ask your institute to investigate, any unauthorized attempts to change authorship or discrepancies in authorship between the submitted and revised versions of your manuscript.

We look forward to receiving your revised manuscript and please do not hesitate to contact us if you have any questions.

Best wishes,

Amanpreet Athwal

on behalf of Janelle Coore
BMC Pregnancy and Childbirth
https://bmcpregnancychildbirth.biomedcentral.com/

Response: Thank you for accepting our manuscript, we have made the necessary amendments.

Assistant Editor Comments:
1. Title page

Please remove the word count from the Title page.

Response: We have removed this.

2. Clean manuscript

At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files. Please ensure that all figures, tables and additional/supplementary files are cited within the text.

Response: We have cleaned the manuscript and have uploaded a clean version of the manuscript.